Linear Algebra Navigator: C. Andy Tsao Class Notes and Homework

Week 3.

Spring 2002.

Outline

We have outline §1.1- §1.2 and done a few examples. This week you will see more examples to give you more firm grasp on what’s going on. Towards the end of Chapter 1, you will see more formal matrix notations. Don’t be intimidated by them. Keep the linear systems and the row operations in mind. You will see that they are essentially the same thing but different expressions.

1. Examples for reducing a general system to row echelon form (REF) and reduced row echelon form (RREF).

2. Distinguish inconsistent system from the consistent ones using their REF. Examples.

Back substitution.

3. Solving simultaneous linear systems with same coefficient matrix but different RHS.

4. Review matrix algebra and highlighting the connection with linear system and Gaussian elimination.

5. For matrix algebra: AB ≠ BA, (AB)^T = B^T A^T . Examples.

6. Idea of proof? Why do we need it anyway? Proofs in Lanew: Idea, why is it true?

Model: Trying to convince a difficult person.

7. Elementray matrices: Equivalent Systems, Elementary matrices (Elementary Row Operations)

Homework (due 020321)

1. §1.1: 8, 9, 10. §1.2: 8, 10. §1.3: 4, 9, 13, 17, 19.

Learning Guide and Announcement

1. Review §1.1–§1.3. Preview §1.4–§1.5.

2. There will be other assignments for next week. Please do at least one problems per day.

On P521–P539 of the textbook, there are answers to selected exercises.

3. The TA’s and their office hours has been announced in our course web. Checked out the recitation schedule. Choose the one suits you better. My office hours are also given there. See you then. ♠♥♦♣

Page 1 of 1 Typeset by AMS-LaTeX